

Author's response to reviews

Title: Environmental exposures and their genetic or environmental contribution to depression and fatigue: a twin study in Sri Lanka

Authors:

Harriet A Ball (harriet.ball@kcl.ac.uk)
Sisira H Siribaddana (nipuna@stmail.lk)
Athula Sumathipala (thula.sumathipala@kcl.ac.uk)
Yulia Kovas (yulia.kovas@kcl.ac.uk)
Nick Glozier (nglozier@usyd.edu.au)
Peter McGuffin (peter.mcguffin@kcl.ac.uk)
Matthew Hotopf (matthew.hotopf@kcl.ac.uk)

Version: 6 **Date:** 30 January 2010

Author's response to reviews:

Changes following formatting:

The manuscript PDF is fine except because you're re-ordered the tables (with the summary table no longer at the end), the following text has been changed to make it a bit more clear:

P15 "(Tables 3 & 4)" changed to (Table 3, and for a summary of findings see Table 4)

p16 ", Tables 4 & 5)" changed to ", Table 5, and for a summary of findings see Table 4).

Thanks, Harriet